Author's response to reviews

Title: Delayed Spinal Extradural hematoma following thoracic spine surgery producing paraplegia: A Case Report

Authors:

Chandra JKB Parthiban (juttyparthiban@hotmail.com)
Shiju A Majeed (shijumajeed@asianetindia.com)

Version: 2 Date: 19 September 2007

Author's response to reviews:

The abstract has been revised with introduction, case presentation and conclusion.

No list of abbreviations has been added. The abbreviation PSEH -- postoperative spinal extradural hematomas is explained in the first line of the text. Rest of the abbreviations used if any are self explanatory.

O/e has been corrected to on examination

Appropriate titles have been added.